

University for the Common Good

Assessing the construct validity of musculoskeletal ultrasound and the rheumatoid arthritis foot disease activity index (RADAI-F5) for managing rheumatoid foot disease Hoque, Anika; Steultjens, Martijn; Dickson, Diane M.; Hendry, Gordon J.

Published in: Rheumatology Advances in Practice

DOI: 10.1093/rap/rkad048

Publication date: 2023

Document Version Publisher's PDF, also known as Version of record

Link to publication in ResearchOnline

Citation for published version (Harvard):

Hoque, A, Steultjens, M, Dickson, DM & Hendry, GJ 2023, 'Assessing the construct validity of musculoskeletal ultrasound and the rheumatoid arthritis foot disease activity index (RADAI-F5) for managing rheumatoid foot disease', *Rheumatology Advances in Practice*, vol. 7, no. 2, rkad048. https://doi.org/10.1093/rap/rkad048

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

If you believe that this document breaches copyright please view our takedown policy at https://edshare.gcu.ac.uk/id/eprint/5179 for details of how to contact us.



## OXFORD

# **Clinical science**

# Assessing the construct validity of musculoskeletal ultrasound and the rheumatoid arthritis foot disease activity index (RADAI-F5) for managing rheumatoid foot disease

# Anika Hogue D<sup>1,\*</sup>, Martijn Steultjens<sup>1</sup>, Diane M. Dickson D<sup>1</sup>, Gordon J. Hendry<sup>1</sup>

<sup>1</sup>Department of Podiatry and Radiography, School of Health and Life Sciences, Glasgow Caledonian University, Glasgow, UK \*Correspondence to: Anika Hogue, Department of Podiatry and Radiography, School of Health and Life Sciences, Glasgow Caledonian University, Cowcaddens Road, Glasgow G4 0BA, UK. E-mail: anika.hogue@gcu.ac.uk

## Abstract

Objective: The RA foot disease activity index (RADAI-F5) is a valid, reliable and clinically feasible patient-reported outcome measure (PROM) for the measurement of RA foot disease activity. Further validation of the RADAI-F5 against musculoskeletal ultrasonography (MSUS) for foot disease activity is necessary before clinical implementation. The aim of this study was to examine the construct validity of the RADAI-F5 in relationship to MSUS and clinical examination.

Methods: Participants with RA completed the RADAI-F5. MSUS was used to evaluate disease activity (synovial hypertrophy/synovitis/tenosynovitis/bursitis) and joint damage (erosion) using greyscale (GS) and power Doppler (PD) at 16 regions in each foot, including joints and soft tissues. These same regions were examined clinically for swelling and tenderness. The construct validity of the RADAI-F5 was assessed using correlation coefficients and a priori-specified hypotheses for the strength of associations.

Results: Of 60 participants, 48 were female, with a mean (s.p.) age of 62.6 (9.96) years and median disease duration of 15.49 (interguartile range 6-20.5) years. Theoretically consistent associations confirming construct validity [95% CI] were observed between the RADAI-F5 and MSUS GS (0.76 [0.57, 0.82]; strong), MSUS PD (0.55 [0.35, 0.71]; moderate), MSUS-detected erosions (0.41 [0.18, 0.61]; moderate), clinical tenderness (0.52 [0.31, 0.68]; moderate) and clinical swelling (0.36 [0.13, 0.55]; weak).

Conclusion: Moderate to strong correlations between RADAI-F5 and MSUS demonstrate the good measurement properties of this instrument. With greater confidence in the utility of the RADAI-F5, clinical use of this new instrument as an adjunct to the disease activity score for 28 joints (DAS-28) could help to identify RA patients at risk for poor functional and radiological outcomes.

## Lay Summary

#### What does this mean for patients?

Rheumatoid arthritis (RA) is a condition that commonly affects the feet. It is important to have a reliable way to measure the activity of RA in the feet. One such measure is called the RA foot disease activity index (RADAI-F5), which is a five-item questionnaire completed by patients. However, the RADAI-F5 still needs to be validated against other methods of assessing foot disease activity, such as musculoskeletal ultrasonography (MSUS) and clinical examination. In this study, 60 participants with RA were enrolled. Participants completed the RADAI-F5 questionnaire. Thereafter, MSUS was used to evaluate disease activity and joint damage in the participants' feet. Clinical examinations to assess swelling and tenderness were also conducted at this appointment. The results showed that there were strong to moderate consistent associations between the RADAI-F5 scores and MSUS greyscale images, MSUS power Doppler and the presence of joint damage. The RADAI-F5 scores were also correlated moderately with clinical tenderness and weakly with clinical swelling. These findings suggest that the RADAI-F5 is a reliable tool for measuring foot disease activity in RA patients. Implementation of the tool in clinical practice might aid in the identification of RA patients who are at risk for poor outcomes and might require more targeted treatment.

Keywords: RA, patient-reported outcome, RADAI-F5, musculoskeletal ultrasound, clinical examination

#### Key messages

- Moderate to strong associations between RADAI-F5 and musculoskeletal ultrasonography provide evidence of the good measurement properties of this instrument.
- RADAI-F5 associations with musculoskeletal ultrasonography-detected greyscale were stronger than clinical assessments of swelling and tenderness.
- The RADAI-F5 is a valuable tool for detecting foot disease activity in patients classified in DAS-28 remission.

Received: 1 February 2023. Accepted: 1 May 2023

© The Author(s) 2023. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

#### Introduction

RA is a chronic systemic inflammatory disease characterized by persistent synovitis with symmetric involvement of peripheral joints. RA often impacts peri-articular structures, such as synovial joints, bursae and tenosynovium [1]. Early diagnosis and an aggressive pharmacological treatment strategy have altered the prognosis for RA patients and placed more individuals into remission [2, 3]. Nevertheless, the prevalence of foot synovitis in RA patients remains high [4]. In 2010, the treatto-target paradigm was proposed, with the therapeutic treatment objective of obtaining clinical remission or low disease activity [5]. This strategy relies on accurate monitoring of disease activity using composite DASs, including the clinical disease activity index (CDAI), the simplified disease activity index (SDAI) and most the most commonly used, DAS-28. Nonetheless, there are various criticisms of the DAS-28, including the subjective nature of clinical assessments of joint tenderness and swelling, the low specificity of the global visual analog scale [6] and its inability to detect foot arthritis [7]. Previous studies reveal that as many as one-third of patients classified in DAS-28 remission present with clinical and/or US-detected manifestations of foot synovitis, which might elevate their risk of radiographic joint degeneration and poor functional outcomes [5, 7]. In addition, although some patients might achieve clinical remission according to the DAS-28, they can still develop foot joint damage in the form of erosions, indicating that management driven by DAS-28 might not address foot disease adequately. Furthermore, a recent study indicated that clinical assessments for swollen and tender foot joints failed to identify the majority of individuals with self-reported foot pain [8]. This study suggests that foot joint counts should be complemented with other self-reporting measures for foot disease [8].

Prior studies have suggested the use of musculoskeletal ultrasonography (MSUS) for monitoring of foot disease in RA as a diagnostic aid to supplement clinical examination [8]. However, MSUS is not routinely used, largely owing to impracticalities, such as training needs and time to perform scans [8, 9]. For self-reporting, several RA foot-specific patient-reported outcome measures (PROMs) have been developed and validated to quantify foot impairments and disability in RA, such as the foot impact scale [10], foot function index [11] and Salford arthritis foot examination instrument [12]. Nonetheless, it appears that these PROMs lack clinical feasibility owing to their lengthiness and associated time burden for completion and scoring [13, 14]. Additionally, these PROMs focus on measurements of foot disability and impairment domains, which are less likely to have clinical utility for a treat-to-target approach with emphasis on early detection of disease activity and intervention to prevent poor outcomes. These limitations led to the development of the RA foot disease activity index (RADAI-F5), a valid, reliable, responsive and clinically feasible five-item PROM for quick and simple self-reporting of the presence and severity of foot disease in RA [13]. The RADAI-F5 was originally evaluated for its construct validity relative to global measures of disease activity, including the DAS-28 ESR and the modified RA disease activity index-5, and disease-specific measures for foot-related impairments and disability (the foot function index and foot impact scale) [13].

Qualitative research to identify barriers and facilitators to implementation of the RADAI-F5 in routine clinical practice suggested that the RADAI-F5 could be used to promote clinician-patient communication, to guide management, to screen patients and to monitor foot disease [14]. However, despite its apparent clinical utility, rheumatologists perceived the demonstration of construct validity of the RADAI-F5 relative to MSUS as crucial before widespread uptake and clinical implementation of this tool [14]. Accordingly, the aim of this study is to evaluate the construct validity of the RADAI-F5 compared with both MSUS and clinical examinations of foot joint and soft tissue tenderness and swelling.

#### Methods

#### Study design

From December 2021 to October 2022, this cross-sectional observational study involved a single visit to the MSUS imaging suite at Glasgow Caledonian University (GCU). Ethical approval was obtained from the North East Research Ethics Committee (21/NE/0130) and GCU Ethics School of Health Subcommittee (HLS/PSWAHS/20/242). Participants were recruited consecutively, and written informed consent was obtained from all participants. Our study conforms to the COnsensus-based Standards for the selection of health Measurement INstruments (COSMIN) Reporting Guideline for studies on measurement properties [15].

#### Study participants

Adults with a diagnosis of RA in accordance with the 2010 revised ACR and EULAR criteria [16, 17] were recruited from National Health Service Greater Glasgow and Clyde rheumatology outpatient clinics at Gartnavel General Hospital, Royal Alexandra Hospital and University Hospital Wishaw. Patients were excluded if they were wheelchair users, had recent foot surgical interventions within the previous 12 months, had received foot/ankle CS injections within the previous 6 months or if they were diagnosed with comorbidities that could have diminished their ability to distinguish between RA-related foot problems and problems attributable to other disease mechanisms, such as diabetes-related peripheral neuropathy. G Power determined that a sample size of 60 participants was needed for adequate detection of a correlation of  $\geq 0.35$  (weak), with a power of 80% and  $\alpha$ -level set at 0.05.

#### Data collection and measures

The participant assessments were all performed on the same day at the Glasgow Caledonian University Human performance laboratory. Demographic and clinical information was collected, including age, sex, disease duration, DAS-28 ESR scores and current medication [18, 19]. The modified version of the RA disease activity index (mRADAI-5) was collected as an additional self-reported measure of global disease activity [20]. Foot disease activity was evaluated using the RADAI-F5. The RADAI-F5 (Supplementary Data S1, available at Rheumatology Advances in Practice online) is a five-item questionnaire completed using a numerical rating scale format from 0 to 10 and scored by calculating an average summary score from the five items, ranging from 0 to 10 [13]. Foot disease remission state was defined as a RADAI-F5 score of  $\leq$ 1.4, and foot disease categories for mild, moderate and high disease activity were defined as: >1.4 to  $\leq 3.45$ , >3.45 to  $\leq$ 5.7, and >5.7, respectively [13].

#### Physical assessment

An independent assessor (G.J.H.) clinically examined 16 regions of each foot, including joints (ankle, subtalar, talonavicular, MTP joints two to five) and soft tissue sites (four intermetatarsal bursae, five plantar metatarsal bursae and the tibialis posterior tendon) for the presence of tenderness and swelling. Given that the questionnaires were locked in a secure locked safe until data collection was complete, G.J.H. completed the clinical evaluations blinded to the MSUS findings and questionnaire outcomes. Tender and swollen joint counts were each scored as present or absent, with a maximum score of 16 for tenderness and swelling for each foot (range 0–32 for both feet).

#### Musculoskeletal ultrasonography

A single postgraduate certificate MSUS-trained podiatrist conducted all MSUS scans. MSUS assessment involved detection and grading of greyscale synovial hypertrophy (GS) and synovitis power Doppler (PD) signals of the same 16 regions of each foot, including joints (ankle, subtalar, talonavicular and MTP joints two to five) and soft tissue sites (four intermetatarsal bursae, five plantar metatarsal bursae and the tibialis posterior tendon). Longitudinal and transverse MSUS scans were performed with a Logiq S8 US machine (GE Medical Systems Ultrasound and Primary Care Diagnostics), using a linear transducer (9-15 MHz). Grading of each region was scored using a semi-quantitative scale of zero to three for GS and PD, as described in Table 1. Erosions and bursae were scored on a dichotomous scale as present or absent. The MSUS score was calculated as the summation of GS and PD scores (range 0-39) and erosions (range 0-7). The principal

3

investigator (A.H.) completing the MSUS evaluation was blinded to the RADAI-F5 scores; however, it was not always possible to obtain an independent examiner to undertake the clinical examination owing to the coronavirus disease 2019 pandemic. As a result, whenever possible, A.H. was blinded to the clinical examination results (n = 44).

#### Statistical analysis

All data were analysed using SPSS v.28 and Excel 2016. Descriptive statistics including age {median [interquartile range (IQR)]}, sex (female:male ratio) and disease duration (median [IQR]) were generated for all participants. The construct validity of the RADAI-F5 was assessed using correlation coefficients and a priori-specified hypotheses for the strength of associations. A priori hypotheses for construct validity included moderate positive correlations between RADAI-F5 scores and clinical examination, clinical tenderness, MSUS GS and PD, and a weak positive correlation with foot erosions. Pearson's correlation and 95% CIs were used to test these hypotheses. Coefficients were interpreted as follows: 0-0.1 = negligible, 0.1-0.39 = weak,0.4 -0.69 =moderate, 0.7 - 0.89 =strong and 0.9 - 1.0 =very strong [26]. Participants were classified to corresponding foot and global disease groups based on RADAI-F5 and DAS-28 values (remission, low, moderate and high). To investigate MSUS-detected foot disease scores further, disease categories were cross-tabulated with MSUS scores for synovial hypertrophy (GS) and PD. This helped to determine the level of active disease in participants who met low disease and remission criteria based on DAS-28 and RADAI-F5 scores. To gain further insight into the behaviour of RADAI-F5 items, item-level

Table 1. Methods of scoring pathology on musculoskeletal ultrasonography

Pathology	Definition	Scoring
Synovial hypertrophy [21]	Abnormal hypoechoic, poorly compressible and non-displaceable intra-articular tissue, which may exhibit power Doppler signal	Score 0: no hypertrophy, independent of presence of effusion Score 1: minimal hypertrophy, with or without effu-
	nuy chinok power Doppler signal	sion up to level of horizontal line connecting bone surfaces
		Score 2: moderate hypertrophy, with or without effu- sion extending beyond joint line but with upper sur- face concave or hypertrophy extending beyond joint line but with upper surface flat
		Score 3: severe hypertrophy, with or without effusion extending beyond joint line but with upper surface convex
Synovitis [21]	Power Doppler interrogation of the synovial tissues	Score 0: no colour flow
		Score 1: $\leq$ 3 colour signals
		Score 2: $<50\%$ of the area filled with colour signals
Erosion [22]	Cortical break seen in longitudinal and transverse	Score 0: absent
	planes measuring >2 mm	Score 1: present
Bursitis [23]	Hypoechoic, well-defined, anechoic or hypoe-	Score 0: absent
	choic, compressible lesion, which may exhibit power Doppler signal	Score 1: present
Tenosynovium hypertrophy [24]	Hypoechoic tendon thickening, with or without fluid in the tendon sheath, which may exhibit	Score 0: normal (i.e. 6 mm in sagittal and 14.1 mm in transverse view)
	power Doppler signals, seen in two perpendicu- lar planes	Score 1: minimal thickening of tendon
		Score 2: moderate thickening of tendon
		Score 3: severe thickening of tendon
Tenosynovitis [25]	Power Doppler interrogation of the tibialis poste-	Score 0: no signals
	rior tendon in two planes	Score 1: signals in only one area of the tendon sheath
		Score 2: signals in more than one area of the widened tendon sheath
		Score 3: signals filling most of the widened tendon sheath

associations with MSUS GS and PD scores were explored using Pearson's correlations.

#### Results

Sixty participants, of whom 80% were female, with a mean age of 62.4 (IOR 50-62) years and median disease duration of 120 months, participated in this study. Forty-three participants were receiving DMARD therapy, and 17 were receiving biologics. For DAS-28 ESR disease categories, 18% were in remission, 28% had low disease activity, 25% had moderate disease activity and 28% had high disease activity. For RADAI-F5 foot disease categories, 10% were in remission, 35% had low foot disease activity, 18% had moderate foot disease activity and 37% had high foot disease activity. Participants typically presented with moderate self-reported foot-related and global disease according to the RADAI-F5 and mRADAI-5 (Table 2). Tender joints and swollen foot joint and soft tissue sites were common, with a median (IQR) of two (four) and eight (one) observed, respectively. B-Mode GS was more prevalent than PD synovitis, with a mean (s.D.) of 14.85 (8.57) compared with 2.75 (3.23). Erosions were less frequently observed, with a mean (S.D.) of 0.70 (1.53).

Almost all of our participants' self-reported the presence of foot or ankle disease 98% (59 of 60), of whom 32 of 60 (53%) were in the moderate/high RADAI-F5 categories. Foot synovial hypertrophy, defined as having a GS score of at least

**Table 2.** Participant descriptive data (n = 60)

Characteristic	All participants $(n = 60)$
Age, years	62.6 [9.97]
Sex, female:male	4:1
Disease duration, years	15.49 [12.19]
DAS-28 ESR	3.83 [1.38]
In remission ( $\leq 2.6$ ), $n$ (%)	11/60 (18)
Low disease (>2.6 to $\leq$ 3.2), <i>n</i> (%)	17/60 (28)
Moderate disease (>3.2 to $\leq$ 5.1), <i>n</i> (%)	15/60 (25)
High disease (>5.1), $n$ (%)	17/60 (28)
Therapy	
DMARDs, <i>n</i> (%)	42/60 (70)
Biologic therapy, $n$ (%)	17/60 (28)
Glucocorticoids between 6 months and 1 year, <i>n</i> (%)	12/60 (20)
None, <i>n</i> (%)	1/60 (2)
mRADAI-5	4.79 [2.05]
In remission ( $\leq 1.4$ ), <i>n</i> (%)	3/60 (5)
Low disease (>1.6 to $\leq$ 3.0), <i>n</i> (%)	9/60 (15)
Moderate disease (>3.2 to $\leq$ 5.4), <i>n</i> (%)	23/60 (38)
High disease ( $>5.6$ ), $n$ (%)	25/60 (42)
RADAI-F5	4.39 [2.69]
In remission ( $\leq$ 1.4), <i>n</i> (%)	6/60 (10)
Low disease (>1.4 to $\leq$ 3.45), <i>n</i> (%)	21/60 (35)
Moderate disease (>3.45 to $\leq$ 5.7), <i>n</i> (%)	11/60 (18)
High disease (>5.7), $n$ (%)	22/60 (37)
Clinical assessments	
TJC, median (IQR)	2 (1-3)
SJC, median (IQR)	8 (0-2)
GS	14.85 [8.57]
PD	2.75 [3.23]
Erosions	0.70 [1.53]

Results are shown as the mean [s.D.], unless specified.

DAS-28: DAS-28 joints; GS: greyscale; IQR: interquartile range; mRADAI-5: modified version of the RA disease activity index; PD: power Doppler; PROMIS-PF: patient-reported outcomes measurement information system—physical function short form questionnaire; RADAI-F5: RA foot disease activity index; SJC: swollen joint count; TJC: tender joint count. two, was prevalent in 57 of 60 (95%) participants. Foot synovitis, defined as having a PD score of at least one, was evident in 38 of 60 (63%) participants. Synovial hypertrophy observed by MSUS was most prevalent in the second and third MTP joints and ankle joint, whereas the tibialis posterior was the most affected soft tissue site. MSUS-detected synovitis was more frequently observed at the ankle joint. The most common sites for clinical detection of swelling were the ankle and subtalar joints, whereas the most common sites for clinical detection of tenderness were the intermetatarsal bursae, the ankle joint and the second and third MTP joints.

# Associations between RADAI-F5 and MSUS and clinical examinations of foot disease

The association between the RADAI-F5 and GS synovial hypertrophy was stronger than expected (Pearsons (r) = 0.75 [95% CI 0.61, 0.84], P < 0.01; Table 3; Fig. 1). As anticipated, a moderate positive association was observed with PD (r = 0.60 [95% CI 0.41, 0.74], P < 0.01) and a weak association with erosions (r = 0.29 [95% CI 0.04, 0.51], P < 0.01). The RADAI-F5 had a weaker than anticipated association with clinical swelling (r = 0.37 [95% CI 0.13, 0.57], P < 0.05) and a moderate association with clinical tenderness (r = 0.44 [95% CI 0.21, 0.62], P < 0.01; Fig. 1). Construct validity was confirmed, with 60% of associations in line with the a priori hypotheses.

# Comparison of RADAI-F5, DAS-28 and MSUS characteristics of the low foot disease/remission foot groups

Seventeen individuals with RA scored positively on the RADAI-F5 despite not having active foot disease as indicated by MSUS. Compared with the moderate-severe RADAI-F5 group, the remission and low disease activity RADAI-F5 group scored significantly higher for item 1 (related to foot disease activity in the previous 6 months), with a mean (S.D.) item score of 2.11 (1.57) and 4.57 (2.28), respectively. Calculating the correlation coefficient for each RADAI-F5 item against MSUS GS and PD findings revealed that the association between RADAI-F5 items and PD exhibited weak positive associations for questions related to foot disease in the last 6 months (item 1) and morning stiffness (item 5). In comparison, GS scores were moderately associated only for questions related to morning stiffness (item 5). Items related to joint tenderness/swelling (item 2), pain (item 3) and foot health (item 4) were strongly associated with GS and moderately associated with PD (Table 4).

Patients in DAS-28 remission scored mean (s.D.) RADAI-F5 scores of 2.67 (2.13). Of the 11 patients classified according to DAS-28 remission, 4 patients remained in remission according to the RADAI-F5, 5 patients were in RADAI-F5 low disease category, and 2 were classified as having high levels of foot disease. The analysis of the DAS-28 categories against MSUS features revealed that 63% of individuals in the DAS-28 remission category had at least grade 2 GS at one or more sites of interest (as seen in Table 5). In comparison, 88% of individuals in the DAS-28 low disease category had at least grade 2 GS, while 100% of those in moderate to high DAS-28 categories had at least grade 2 GS. Furthermore, 54% of those in the low DAS-28 category still had PD signals in the foot structures of interest (as seen in Table 5).

#### Management of rheumatoid foot disease

Table 3. Pearson's correlations of RA foot disease activity index with objective measures of foot disease activity for a priori hypothesis testing

Measure $(n = 60)$	Correlation coefficient (95% CI)	Strength	A priori hypothesis
Clinical swelling	0.37 (0.13, 0.57)	Weak	Moderate
Clinical tenderness	0.44 (0.21, 0.62)	Moderate	Moderate
Synovial hypertrophy (GS)	0.75 (0.61, 0.84)	Strong	Moderate
Synovitis (PD)	0.60(0.41, 0.74)	Moderate	Moderate
Erosions	0.29 (0.04, 0.51)	Weak	Weak

GS: greyscale; PD: power Doppler.



Figure 1. Associations of the RA foot disease activity index with synovial hypertrophy (A), power Doppler (B), clinical swelling (C) and clinical tenderness (D) RADAI-F5: RA foot disease activity index

Table 4. RA foot disease activity index item associations with musculoskeletal ultrasonography

RADAI-F5 Items	GS MSUS [correlation	Strength	PD MSUS [correlation	Strength
	coefficient (95% CI)] <sup>2</sup>	coefficient (95% CI)]"		
Item 1	0.62 (0.44, 0.76)	Moderate	0.45 (0.22, 0.63)	Moderate
Item 2	0.75 (0.61, 0.84)	Strong	0.57 (0.37, 0.71)	Moderate
Item 3	0.73 (0.59, 0.83)	Strong	0.58 (0.38, 0.73)	Moderate
Item 4	0.68 (0.52, 0.80)	Moderate	0.62(0.44, 0.76)	Moderate
Item 5	0.59 (0.40, 0.73)	Moderate	0.44 (0.21, 0.62)	Moderate

<sup>a</sup> Pearson's correlations, all significant at P < 0.05. GS: greyscale; MSUS: musculoskeletal ultrasonography; PD: power Doppler.

Table 5. DAS-28 disease category summary statistics

DAS-28	GS MSUS $\geq$ grade 2 [ $n$ (% affected)]	PD MSUS $\geq$ grade 1 [ $n$ (% affected)]	MSUS erosion at more than one site [ <i>n</i> (% affected)]
In remission $(n = 11)$	7 (64)	6 (54)	0 (0)
Low $(n = 17)$	15 (88)	9 (53)	3 (18)
Moderate $(n = 15)$	15 (100)	9 (60)	6 (40)
High $(n = 17)$	17 (100)	13 (77)	11 (65)

GS: greyscale; MSUS: musculoskeletal ultrasonography; PD: power Doppler.

## Discussion

To our knowledge, this is the first study to use the RADAI-F5 to examine the relationship between MSUS-detected and selfreported foot disease activity. The RADAI-F5 demonstrates good construct validity, in line with a priori expectations, suggesting that this new tool has moderate to strong associations with MSUS GS, PD and erosions. This work provides additional evidence to our previous validation work [13]. The DAS-28 classed n = 11 participants in remission; of these, 54% presented with at least grade 1 PD affecting at least one site of interest. In addition, n = 17 individuals were assigned to the DAS-28 low disease category, and 53% presented with at least grade 1 PD affecting at least one site of interest within the foot. Nevertheless, this result was anticipated based on evidence that composite disease activity indices that exclude foot joints might not represent foot synovitis accurately [13]. Consequently, systemic therapy escalated solely on the basis of DAS-28 values and in the context of infrequent foot examinations during routine consultations might result in persistent foot disease and suboptimal management. Our findings demonstrate that the RADAI-F5 has clinical utility in detecting the activity of inflammatory foot disease in individuals with established and early RA, and that monitoring the feet is crucial for these patients owing to the high incidence of foot involvement in RA. The incidence of foot involvement in RA has been described extensively in the literature, with earlier estimates ranging from 56 to 100% [13, 27]. Despite advancements in pharmaceutical management, the prevalence of foot disease remains high in our sample, as evidenced by the 95% of participants who exhibited MSUS-detected GS in the foot and 64.4% who had active PD signals in the foot or ankle.

In clinical practice, people with RA are routinely evaluated by the number of swollen and tender joints to determine disease activity. The addition of a 12-joint foot count to the DAS-28 in a foot or ankle pain detection study showed that 92.3% and 61.4% of patients with foot pain had no swollen or tender foot joints, respectively, revealing a significant discrepancy between clinical examination and self-reported foot pain [13]. The authors speculated that the high incidence of foot pain might be attributable to structures not included in the 12-joint foot count, such as the midfoot and other soft tissue structures. Despite this, the vast majority of our patients did not test positive for oedema, even when soft tissue structures and hindfoot joints were included in the assessment. This suggests that clinical examination undertaken in isolation, without additional MSUS imaging and/or a valid method of foot disease self-report, might be a suboptimal approach. Incorporating the RADAI-F5 as an adjunct to the clinical examination might assist in the identification of more individuals with active foot disease.

Several individuals (n = 17) scored higher than zero on the RADAI-F5 despite the absence of synovitis, tenosynovitis or erosions in their foot and ankle joints. Those in remission and low RADAI-F5 disease categories scored the highest on item 1 (related to foot disease activity in the previous 6 months), indicating that there might have been active disease in the past that was not present on the day of the MSUS scan and residual foot symptoms attributed to previous active foot disease. Despite the absence of active Doppler signals, the participants in the high RADAI-F5 disease group appeared to have GS scores typically grade 2 or 3, indicating that the presence of synovial hypertrophy is strongly associated with selfperceptions of foot disease. Typically, the primary focus of pharmacological treatment for RA is the inflammatory component of the disease, which is identified by Doppler activity rather than synovial hypertrophy [28, 29]. Witt et al. [30] found that grade 1 synovial hypertrophy is evident in healthy people and is unresponsive to therapy in early and established RA, and Padovano et al. [31] identified similar findings in several healthy controls with grade 1 synovial hypertrophy. Nonetheless, Terslev et al. [32] reported that grade 1 synovial hypertrophy could improve with the initiation of biological treatment, regardless of the absence of Doppler activity. Similar outcomes have been observed for tenosynovitis, as grade 1 tenosynovitis without positive Doppler activity improves with therapy [33]. Moreover, synovial hypertrophy in RA patients has been associated with an early recurrence of inflammatory arthritis and is predictive of erosion progression [34, 35]. This indicates that GS synovial hypertrophy is still meaningful and responsive to change and that eradication of Doppler signals might not be the main therapeutic aim in RA patients when contemplating therapy escalation. In accordance with OMERACT [21], synovial hypertrophy without Doppler activity is a symptom of active disease and should be considered when assessing the activity of foot disease using MSUS. As such, our findings might indicate that synovial hypertrophy still impacts upon self-perceptions of foot disease. Strong associations between GS and RADAI-F5 are therefore encouraging, because they suggest that individuals with RA might be able to detect localized inflammatory alterations in synovial hypertrophy and indicate the possible use of RADAI-F5 as a screening tool.

It is noteworthy that the RADAI-F5 does not specify which areas of the foot are affected; rather, it offers a comprehensive score for foot disease. Use of the RADAI-F5 in podiatry and rheumatology clinics has the potential to enhance the patient experience and quality of care by facilitating early detection of RA-related foot disease and informing therapeutic approaches based on RADAI-F5 disease classification categories [13]. We suggest that patients with RA be offered the opportunity to complete the RADAI-F5 in the waiting area before their appointment. We recommend providing patients in RADAI-F5 remission and with low disease with verbal and written information regarding their condition and its management, in addition to footwear guidance and, if needed, functional orthotics. A RADAI-F5 score in the moderate and high categories should prompt additional inquiry by clinical examination of foot joints and soft tissues, or patients can complete a foot map, in which they shade problematic areas. If clinical examination verifies the existence of suspicious joints, we suggest considering MSUS imaging to confirm the presence of synovitis, in which case CS injections would be recommended. In addition, allied health-care providers should consider referral to rheumatology for possible medication escalation. The RADAI-F5 can also be used in remote consultations to identify patients who necessitate immediate inperson appointments, where clinical or MSUS assessments are advisable.

There are some limitations to this study. Firstly, quantifying the symptomatic impact of subclinical foot synovitis is difficult; thus, longitudinal follow-up is required. Specifically, longitudinal data following pharmacological treatment are necessary to evaluate the relative responsiveness of MSUS to establish whether the RADAI-F5 can evaluate the effectiveness of therapy in clinical care. Secondly, there is a risk of selection bias because individuals who participated in the research were recruited from a rheumatology outpatient clinic, who might have experienced a more significant degree of foot involvement than patients in DAS-28 remission on or off medication, who would be less likely to attend as frequently. Another limitation of this study is that the MSUS examiner was not blinded to all clinical foot examinations, thus investigator bias cannot be discounted. Nonetheless, given that the RADAI-F5 scores were concealed from the lead investigator, it is unlikely that the association between RADAI-F5 scores and MSUS-detected foot disease was compromised.

The present study found that RA foot and ankle disease remains prevalent even in patients in DAS-28 remission. As anticipated, the correlation between RADAI-F5 scores and MSUS-detected synovial hypertrophy was more significant than clinical assessments of joint swelling and tenderness. With greater confidence in the utility of the RADAI-F5, clinical use of this new instrument could help to identify RA patients at risk for poor functional and radiological outcomes. Adopting the RADAI-F5 as an adjunct to composite disease activity indices, such as the DAS-28, might improve local disease detection and guide new foot care protocols and has potential for use in a wide range of clinical applications, including triage, improved communication with patients and multidisciplinary teams, and remote consultations.

#### **Supplementary material**

Supplementary material is available at *Rheumatology Advances in Practice* online.

#### **Data availability**

The data underlying this article cannot be shared publicly due to the privacy of individuals that participated in the study. The data will be shared on reasonable request to the corresponding author.

### Funding

No specific funding was received from any bodies in the public, commercial or not-for-profit sectors to carry out the work described in this manuscript.

*Disclosure statement*: The authors have declared no conflicts of interest.

We are grateful to the participants of this study for their time and willingness to share their experiences, and to Professor Duncan Porter, Dr James Dale, Dr Elizabeth Clarke and Dr Derek Baxter for their assistance in participant recruitment. We are grateful to Rona Baille and Kathleen Cowan for sending participant contact information to the principal investigator.

#### References

- Guo Q, Wang Y, Xu D *et al.* Rheumatoid arthritis: pathological mechanisms and modern pharmacologic therapies. Bone Res 2018; 6:15–4.
- Curtis JR, Singh JA. Use of biologics in rheumatoid arthritis: current and emerging paradigms of care. Clin Ther 2011;33:679–707.
- Richter MD, Crowson CS, Matteson EL, Makol A. Orthopedic surgery among patients with rheumatoid arthritis: a populationbased study to identify risk factors, sex differences, and time trends. Arthritis Care Res 2018;70:1546–50.
- Wilson O, Hewlett S, Woodburn J, Pollock J, Kirwan J. Prevalence, impact and care of foot problems in people with rheumatoid arthritis: results from a United Kingdom based cross-sectional survey. J Foot Ankle Res 2017;10:46.
- Ramiro S, Landewé RB, van der Heijde D *et al*. Is treat-to-target really working in rheumatoid arthritis? A longitudinal analysis of a cohort of patients treated in daily practice (RA BIODAM). Ann Rheum Dis 2020;79:453–9.
- Rocha MOC, Cruz AACF, Santos DO *et al.* Sensitivity and specificity of assessment scales of dentin hypersensitivity – an accuracy study. Brazilian Oral Res 2020;34:43.
- Ciurtin C, Jones A, Brown G *et al.* 063 Real-life benefits of ultrasound evaluation of hand and foot synovitis and lack of correlation with DAS-28 in rheumatoid arthritis. Rheumatology 2019;58: 6345–54.
- Simonsen MB, Hørslev-Petersen K, Cöster MC, Jensen C, Bremander A. Foot and ankle problems in patients with rheumatoid arthritis in 2019: still an important issue. ACR Open Rheumatol 2021;3:396–402.
- Otter SJ, Lucas K, Springett K *et al.* Foot pain in rheumatoid arthritis prevalence, risk factors and management: an epidemiological study. Clin Rheumatol 2010;29:255–71.
- Helliwell P, Reay N, Gilworth G et al. Development of a foot impact scale for rheumatoid arthritis. Arthritis Rheum 2005;53: 418-22.
- Budiman-Mak E, Conrad KJ, Mazza J, Stuck RM. A review of the foot function index and the foot function index – revised. J Foot Ankle Res 2013;6:5.
- Walmsley S, Williams A, Ravey M, Graham A. Development of the Salford Rheumatoid Arthritis Foot Evaluation Instrument (SAFE). J Foot Ankle Res 2013;6:34.
- 13. Hoque A, Gallagher K, McEntegart A *et al.* Measuring inflammatory foot disease in rheumatoid arthritis: development and validation of the rheumatoid arthritis foot disease activity index–5. Arthritis Care Res 2021;73:1290–9.
- 14. Hoque A, Steultjens M, Dickson DM, Hendry GJ. Patients' and clinicians' perspectives on the clinical utility of the Rheumatoid Arthritis Foot Disease Activity Index. Rheumatol Int 2022;42: 1807–17.
- Mokkink LB, de Vet HCW, Prinsen CAC *et al.* COSMIN Risk of Bias checklist for systematic reviews of Patient-Reported Outcome Measures. Qual Life Res 2018;27:1171–9.
- Aletaha D, Neogi T, Silman AJ et al. 2010 Rheumatoid arthritis classification criteria: an American College of Rheumatology/European League Against Rheumatism. Arthritis Rheum 2010;62:2569–81.
- 17. Kay J, Upchurch KS. ACR/EULAR 2010 rheumatoid arthritis classification criteria. Rheumatology (Oxford) 2012;51:vi5–9.

- Strait A, Castillo F, Choden S *et al.* Demographic characteristics of participants in rheumatoid arthritis randomized clinical trials: a systematic review. JAMA Network Open 2019;2:e1914745.
- 19. Ataman S, Sunar I, Bodur H *et al*. Demographic and clinical characteristics of patients with sustained and switching treatments using biological and targeted synthetic disease-modifying antirheumatic drugs: a multicenter, observational cross-sectional study for rheumatoid arthritis. Rheumatol Ther 2022;9:223–41.
- Leeb BF, Haindl PM, Maktari A, Nothnagl T, Rintelen B. Patientcentered rheumatoid arthritis disease activity assessment by a modified RADAI. J Rheumatol 2008;35:1294–9.
- 21. D'Agostino M, Terslev L, Aegerter P *et al.* Scoring ultrasound synovitis in rheumatoid arthritis: a EULAR-OMERACT ultrasound taskforce—Part 1: definition and development of a standardised, consensus-based scoring system. RMD Open 2017;3:e000428.
- 22. Grosse J, Allado E, Albuisson É *et al.* Evaluation of bone erosions in rheumatoid arthritis: the ultrasound score for erosions versus the modified Sharp/van der Heijde score for erosions. J Rheumatol 2021;48:335–8.
- 23. do Prado AD, Staub HL, Bisi MC *et al.* Ultrasound and its clinical use in rheumatoid arthritis: where do we stand? Adv Rheumatol 2018;58:19.
- 24. Naredo E, D'Agostino MA, Wakefield RJ *et al.*; OMERACT Ultrasound Task Force. Reliability of a consensus-based ultrasound score for tenosynovitis in rheumatoid arthritis. Ann Rheum Dis 2013;72:1328–34.
- Swen N, Brown AK, Schmidt WA *et al.* Musculoskeletal ultrasound including definitions for ultrasonographic pathology. J Rheumatol 2005;32:2485–7.
- Schober P, Boer C, Schwarte LA. Correlation coefficients: appropriate use and interpretation. Anesth Analg 2018;126: 1763–8.
- 27. Rome K, Gow PJ, Dalbeth N, Chapman JM. Clinical audit of foot problems in patients with rheumatoid arthritis treated at Counties

Manukau District Health Board, Auckland, New Zealand. J Foot Ankle Res 2009;2:16.

- 28. Wechalekar MD, Lester S, Hill CL *et al.* Active foot synovitis in patients with rheumatoid arthritis: unstable remission status, radiographic progression, and worse functional outcomes in patients with foot synovitis in apparent remission. Arthritis Care Res 2016; 68:1616–23.
- Colebatch AN, Edwards CJ, Østergaard M et al. EULAR recommendations for the use of imaging of the joints in the clinical management of rheumatoid arthritis. Ann Rheum Dis 2013;72:804–14.
- Witt M, Mueller F, Nigg A *et al.* Relevance of grade 1 gray-scale ultrasound findings in wrists and small joints to the assessment of subclinical synovitis in rheumatoid arthritis. Arthritis Rheum 2013;65:1694–701.
- Padovano I, Costantino F, Breban M, D'Agostino MA. Prevalence of ultrasound synovial inflammatory findings in healthy subjects. Ann Rheum Dis 2016;75:1819–23.
- 32. Terslev L, Østergaard M, Sexton J, Hammer HB. Is synovial hypertrophy without Doppler activity sensitive to change? Post-hoc analysis from a rheumatoid arthritis ultrasound study. Arthritis Res Ther 2018;20:224.
- 33. Ammitzbøll-Danielsen M, Østergaard M, Naredo E, Terslev L. Validity and sensitivity to change of the semi-quantitative OMERACT ultrasound scoring system for tenosynovitis in patients with rheumatoid arthritis. Rheumatology (Oxford) 2016;55:2156–66.
- Scirè CA, Montecucco C, Codullo V et al. Ultrasonographic evaluation of joint involvement in early rheumatoid arthritis in clinical remission: power Doppler signal predicts short-term relapse. Rheumatology (Oxford) 2009;48:1092–7.
- 35. Saleem B, Brown AK, Keen H *et al.* Disease remission state in patients treated with the combination of tumor necrosis factor blockade and methotrexate or with disease-modifying antirheumatic drugs: a clinical and imaging comparative study. Arthritis Rheum 2009;60:1915–22.



# A 2nd generation, JAK1 preferential inhibitor for moderate to severe RA<sup>1-6</sup>

While 1st generation JAK inhibitors are relatively non-selective,<sup>2-6</sup> JYSELECA has over 5x greater potency for JAK1 over JAK2/3 and TYK21\*

Balancing sustained efficacy<sup>7-11</sup> with acceptable tolerability<sup>1,12</sup>



( )

Indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti-rheumatic drugs.<sup>1</sup> May be used as monotherapy or in combination with methotrexate.<sup>1</sup>

\*From biochemical assays, the clinical relevance of which is uncertain. JAK, Janus kinase; RA, rheumatoid arthritis; TYK, tyrosine kinase.

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information.

۲

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information. **JYSELECN** figotinib 100 mg or 200 mg film-coated tablets. **Indication:** Jyseleca is indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti rheumatic drugs (DMARDs). Jyseleca may be used as monotherapy or in combination with methotrexate (MTX). **Dosage:** <u>Adults</u>; 200 mg once daily. Taken orally with/without food. It is recommended that tablets are swallowed whole. <u>Laboratory Monitoring:</u> Refer to the SmPC for information regarding laboratory monitoring and dose initiation or interruption. <u>Elderly</u>, 4 starting dose of 100 mg of filgotinib once daily is recommended for patients aged 75 years and older as clinical experience is limited. <u>Renal impairment</u>: No dose adjustment required in patients with estimated creatinine clearance (CrCl) ≥ 60 mL/min. A dose of 100 mg of filgotinib once daily is recommended for patients with estimated and by is recommended for patients. Safety and efficacy not yet established. **Contraindications**: Hypersensitivity to the active substance or to any of the excipients. Active tuberculosis (TB) or active serious infections. Pregnancy. **Warnings/Precautions**: See SmPC for full information. <u>Immunosuppression</u>: combination use, with immunosuppressints (AK) inhibitors is not recommended as a risk of additive immunosuppressions infections such as pneumonia and opportunistic infections equipations: thypersensitivity to the active sub excluded. <u>Infections</u>; Infections, including serious infections, end additive immunosuppressions index (TR) indications is the assessed prior to initiating in patients with risk factors for infections (see SmPC). Patients should be closely monitored for the development of initiating in patients with restination of restruptions. have been reported, Kisk benefit should be assessed phore of nitating in patients with risk factors for infections (see SmPC). Vatients should be closely monitored for the development of igns and symptoms of infections during and after fligotinib reatment. Treatment should be interrupted if the patient

is not responding to antimicrobial therapy, until infection is controlled. There is a higher incidence of serious infections in the elderly aged 75 years and older, caution should be used when treating this population. <u>Tuberculosis</u>, Patients should be screened for TB before initiating filgotinib, and filgotinib should not be administered to patients with active TE. <u>Viral</u> <u>reactivation</u>: Cases of herpes virus reactivation (e.g., herpes zoster), were reported in clinical studies (see SmPC). If a patient develops herpes zoster, filgotinib treatment should be temporarily interrupted until the episode resolves. Screening for viral hepatitis and monitoring for reactivation should be performed. <u>Malignancy</u>: Immunomodulatory medicinal products may increase the risk of malignancies. Malignancies were observed in clinical studies (see SmPC). <u>Ferlility</u>. In animal studies, decreased ferlility, impaired spermatogenesis, and histopathological effects on male reproductive organs were observed (see SmPC). The potential effect of filgotinib on sperm production and male fertility in humans is currently unknown. <u>Haematological abnormalities</u>; Do not start therapy, or temporarily stop, if Absolute Neutrophil Count (ANC) 1< 10° (cells/L, ALC - OS + 10° cells/L or chaemoglobin «B g/dL. Temporarily stop therapy if these values are observed during routine patient management. <u>Vaccinations</u>; Use of live vaccines during, or immediately prior to, filgotinib treatment is not recommended. <u>Lipids</u>: Treatment with filgotinib was associated with dose dependent increases in lipid parameters, including total cholesterol, and high-density lipoprotein (LDL) levels, while low density lipoprotein (LDL) levels were slightly increased (see SmPC). <u>Cardiovascular</u> *tisk*; Rheumatoid arthritis patients have an increased risk for cardiovascular disorders. Patients should have risk factors (e.g., hypertension, hyperlipidaemia) managed as part of usual standard of care. <u>Venous thrombobembolism</u>: Events of deep venous thrombosis (OVT) and pulmona of DVT/PE, or patients undergoing surgery, and prolonged

#### Learn more at strengthofbalance.co.uk

۲

immobilisation. Lactose content: Contains lactose; patients with rare hereditary problems of galactose intolerance, total lactase deficiency or glucose-galactose malabsorption should not take filgotinib. **Pregnancy/Lactation**: Filgotinib is contraindicated in pregnancy. Filgotinib should not be used during breast-feeding. Women of childbearing potential must use effective contraception during and for at least 1 week after cessation of treatment. **Driving/Using machinery**: No or negligible influence, however dizziness has been reported. **Side effects:** See SmPC for full information. <u>Common (21/100)</u>: herpes zoster, pneumonia, neutropenia, hypercholesterolasemia infection and dizziness. <u>Uncommon (s1/1000 to 1/100)</u>; herpes zoster, pneumonia, neutropenia, hypercholesterolaemia and blood creatine phosphokinase increase. Serious side effects: See SmPC for full information **Legal category**: POM **Pack**: 30 film-coated tablets/bottle **Price**: UK Basic NHS cost: £863.10 Marketing authorisation number(s): <u>Great Britain</u> Jyseleca 100mg film-coated tablets PLGB 42147/0002 hypeleca 200mg film-coated tablets PLGB 42147/0002 Northern Ireland Jyseleca 100mg film-coated tablets EU/1/20/1480/001 EU/1/20/1480/003 Jyseleca 200mg film-coated tablets EU/1/20/1480/003 EU/1/20/1480/004 **Further information**: Galapagos UK, Belmont House, 148 Belmont Road, Uxbridge (DB8 105, United Kingdom 00800 7387 1345 **medicalinfo@glgg**. <u>com</u> Jyseleca<sup>®</sup> is a trademark. **Date of Preparation**: January 2022 UK-RA-FIL-202201-00019 **W** Additional monitoring required Additional monitoring required

Adverse events should be reported. For Great Britain and Northern Ireland, reporting form and information can be found at <u>yellowcard.mhra.gov.</u> and information can be found at <u>yellowcard.mnra.gov.u</u> or via the Yellow Card app (download from the Apple Ap Store or Google Play Store). Adverse events should also be reported to Galapagos via email to DrugSafety.UK.Ireland@glpg.com or 00800 7878 1345

References: 1. JYSELECA SPC. Available at: www.medicines.org.uk. Last accessed: June 2022. 2. Angelini J, et al. Biomolecules 2020;10(7):E1002. 3. Banerjee S, et al. Drugs 2017;77:521–546. 4. O'Shea JJ, et al. Nat Rev Rheumatol 2013;9(3):173–182. 5. Traves PG, et al. Ann Rheum Dis 2021;0(1-11. 6. McInnes IB, et al. Arthr Res Ther 2019;21:183. 7. Combe B, et al. Ann Rheum Dis 2021;doi:10.1136/ annrheumdis-2020-219214. 8. Genovese MC, et al. JAMA 2019;322 (4):315–325. 9. Westhowers R, et al. Ann Rheum Dis 2021;doi:10.1136/annrheumdis-2020-219213. 10. Combe B, et al. Arthritis Rheumatol 2021;73(suppl 10). https://acrabstract/clinical-ou-week-48-of-fig0tinib-treatment-in-an-ongoing-long-term-extension-trial-of-a-plates-twith-inadequate-response-to-mtx-initially-treated-with-filgotinib-ra-adalimumab-during-th/. Last accessed: June 2022. 11. Buch MH, et al. Arthritis Rheumatol 2021;73 (suppl 10). https://acrabstracts.org/abstract/clinical-outcomes-up-to-week-48-of-nogoing-filgotinib-tra-abstracts.org/abstract/Jactastracts.org/abstract/sorg/abstract/linical-outcomes-up-to-week-48-of-Ingoing-filgotinib-tra-abstracts.org/abstract/linical-outcomes-up-to-week-48-of-nogoing-filgotinib-tra-abstracts.org/abstract/linical-outcomes-up-to-week-48-of-ongoing-filgotinib-tra-abstracts.org/abstract/lineg/abstract/linegrated-safety-analysis-update-for-filgotinib-in-patients-with-moderately-to-severely-active-rheumatoid-arthritis-receiving-treatment-over-a-median-of-2-2-years/. Last accessed: June 2022.

۲



June 2022 GB-RA-JY-202205-00033

JYSELECA, GALAPAGOS and the JYSELECA and GALAPAGOS logos are registered trademarks of Galapagos NV. © 2022 Galapagos NV. All rights reserved.

13/06/2022 14:24